Author's response to reviews

Title: The Creation of New Rotation Arc to the Rat Latissimus Dorsi Musculo-cutaneous Flap with Delay Procedures

Authors:

Dr Eray Copcu (copcu@lycos.com)
Nazan Sivrioglu (nsivrioglu@adu.edu.tr)
Alper Aktas (aaktas@msn.com)
Yucel Oztan (yuceloztan@yahoo.com)

Version: Date: 4 Dec 2003

Dear Editor,

We revised our manuscript according to the comments which listed E-Mail dated November 26, 2003. All revisions were listed below. I declare that the co-author is agreed on the revision.

We hope you will find the revised manuscript satisfactory. If any further changes are necessary, we would be happy to improve.

Yours sincerely,

Eray COPCU

Revisions:

1. Title: Unnecessary capitalisation were removed
2. Competing Interests were included as "none declared"
3. Author Contributorship was included at the end of the text
4. Table 3 was submitted as a Figure (figure 10)
5. Figures were revised and all figures were numbered as 1, 2, .., 10. Text were revised according to the new numbers of the figures.